Author's response to reviews

Title: A randomised non-inferiority controlled trial of a single versus a four intradermal sterile water injection technique for relief of continuous lower back pain during labour.

Authors:

Nigel Lee (nigel.lee@mater.org.au)
Peter Coxeter (pcoxeter@mmri.mater.org.au)
Michael Beckmann (michael.beckmann@mater.org.au)
Joan Webster (joan_webster@health.qld.gov.au)
Vanessa Wright (vanessa.wright@mater.org.au)
Tric Smith (tric_smith@health.qld.gov.au)
Sue Kildea (sue.kildea@mater.org.au)

Version: 3 Date: 12 February 2011

Author's response to reviews: see over
Re: Subject: MS: 2304705914968803 - A randomised non-inferiority controlled trial of a single versus a four intradermal sterile water injection technique for relief of continuous lower back pain during labour.

Thankyou for your email advice regarding the research protocol.

In response to the revisions detailed below, we do have consent from the persons portrayed in Figures 1 and 2 for their images to be used and this has been noted in the manuscript.

The flowchart (Figure 3) has been amended to show the number of participants in each arm.

The amended manuscript and EPS file have been uploaded as requested.

Kind Regards

Nigel Lee

Requested revisions:
1) Do you have consent to publish the pictures of the participants in Figures 1 and 2. Please clarify this and document that consent to publish was received in the manuscript.

2) Please complete the flowchart in Figure 3 so that it shows the numbers of participants in each arm.